Author’s response to reviews

Title: Prevalence and predictors of overweight and obesity among school-aged children in urban Ghana

Authors:
Richmond Aryeetey (raryeetey@ug.edu.gh)
Anna Lartey (aalartey@hotmail.com)
Grace Marquis (grace.marquis@mcgill.ca)
Helena Nti (helenanti@ymail.com)
Esi Colecraft (colecraft_s@hotmail.com)
Patricia Brown (kwakaib@hotmail.com)

Version: 2 Date: 29 Sep 2017

Author’s response to reviews:

Editor Comments:

At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

Response:

a clean version without any highlighting is being submitted.

Before the conclusion section, the authors need to add a separate limitation paragraph in addition to the previous strength paragraph, describing the major limitations of the manuscript: including but not limited to the cross-sectional nature of the study design, the use of a very crude dietary assessment tool etc.
Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Instructions for Authors on the journal homepage.

Response

the manuscript has been revised to conform to the Journal's format, as requested

Response:

A separate limitations paragraph discussing limitations of the study have been included on page 14, lines 324-331.